Author’s response to reviews

Title: Colonic Tuberculosis mimicking Crohn's disease; Report of a case

Authors:

Constantinos Chatzicostas (matrix01gr@yahoo.gr)
Ioannis E Koutroubakis (ktjohn@her.forthnet.gr)
Maria Tzardi (tzardi_maria@yahoo.co.uk)
Maria Roussomoustakaki (rousso@med.uoc.gr)
Panagiotis Prassopoulos (gournick@med.uoc.gr)
Elias A Kouroumalis (kouroum@med.uoc.gr)

Version: 2 Date: 19 Apr 2002

Reply point by point to the reviewers
A. (Prof S.P. Misra)
1. We tried to improve English in the revised version
2. The conclusion has been changed according to the suggestion (page 8, line 3)
3. All the proposed corrections of mistakes have been done
4. The duration of follow up is now mentioned(page 5, line 18)
5. The references have been arranged in the Vancouver style

B.(Dr S Odes)
Thank you